UNIVERSITY^{OF} BIRMINGHAM University of Birmingham Research at Birmingham

Health system assessment for access to care after injury in low- or middle-income countries

Whitaker, John; Edem , Idara ; Togun, Ella; Amoah , Abena S. ; Dube , Albert ; Chirwa , Lindani ; Munthali , Boston ; Brunelli, Giulia ; Van Boeckel, Thomas ; Rickard, Rory ; Leather , Andrew JM ; Davies, Justine

DOI: 10.1371/journal.pmed.1004344

License: Creative Commons: Attribution (CC BY)

Document Version Publisher's PDF, also known as Version of record

Citation for published version (Harvard):

Whitaker, J, Edem , I, Togun, E, Amoah , AS, Dube , A, Chirwa , L, Munthali , B, Brunelli, G, Van Boeckel, T, Rickard, R, Leather , AJM & Davies, J 2024, 'Health system assessment for access to care after injury in low- or middle-income countries: A mixed methods study from Northern Malawi', *PLoS Medicine*, vol. 21, no. 1, e1004344. https://doi.org/10.1371/journal.pmed.1004344

Link to publication on Research at Birmingham portal

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

•Users may freely distribute the URL that is used to identify this publication.

•Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.

•User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?) •Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.



Citation: Whitaker J, Edem I, Togun E, Amoah AS, Dube A, Chirwa L, et al. (2024) Health system assessment for access to care after injury in lowor middle-income countries: A mixed methods study from Northern Malawi. PLoS Med 21(1): e1004344. https://doi.org/10.1371/journal. pmed.1004344

Academic Editor: Jennifer Thorley, N/A, UNITED KINGDOM

Received: October 27, 2022

Accepted: January 10, 2024

Published: January 22, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1004344

Copyright: © 2024 Whitaker et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data relevant to this report's analysis are within the manuscript and supporting information files.

RESEARCH ARTICLE

Health system assessment for access to care after injury in low- or middle-income countries: A mixed methods study from Northern Malawi

John Whitaker^{1,2,3}*, Idara Edem^{2,4,5}, Ella Togun², Abena S. Amoah^{6,7}, Albert Dube⁶, Lindani Chirwa^{8,9}, Boston Munthali^{10,11}, Giulia Brunelli¹², Thomas Van Boeckel^{12,13}, Rory Rickard³, Andrew JM Leather^{2‡}, Justine Davies^{1,14,15‡}

1 Institute of Applied Health Research, University of Birmingham, Birmingham, United Kingdom, 2 King's Centre for Global Health and Health Partnerships, School of Life Course and Population Sciences, Faculty of Life Sciences and Medicine, King's College London, London, United Kingdom, 3 Academic Department of Military Surgery and Trauma, Royal Centre for Defence Medicine, Birmingham, United Kingdom, 4 Insight Institute of Neurosurgery & Neuroscience, Flint, Michigan, United States of America, 5 Michigan State University, East Lansing, Michigan, United States of America, 6 Malawi Epidemiology and Intervention Research Unit (formerly Karonga Prevention Study), Chilumba, Malawi, 7 Faculty of Epidemiology and Population Health, London School of Hygiene & Tropical Medicine, London, United Kingdom, 8 Karonga District Hospital, Karonga District Health Office, Karonga, Malawi, 9 School of Medicine & Oral Health, Department of Pathology, Kamuzu University of Health Sciences (KUHeS), Blantyre, Malawi, 10 Mzuzu Central Hospital, Department of Orthopaedic Surgery, Mzuzu, Malawi, 11 Lilongwe Institute of Orthopaedic and Neurosurgery, Lilongwe, Malawi, 12 Health Geography and Policy Group, ETH Zurich, Zurich, Switzerland, 13 Center for Disease Dynamics Economics and Policy, Washington, DC, United States of America, 14 Medical Research Council/Wits University Rural Public Health and Health Transitions Research Unit, Faculty of Health Sciences, School of Public Health, University of the Witwatersrand, Johannesburg, South Africa, 15 Department of Global Surgery, Stellenbosch University, Stellenbosch, South Africa

‡ These authors are joint senior authors on this work. * j.whitaker@bham.ac.uk

Abstract

Background

Injuries represent a vast and relatively neglected burden of disease affecting low- and middle-income countries (LMICs). While many health systems underperform in treating injured patients, most assessments have not considered the whole system. We integrated findings from 9 methods using a 3 delays approach (delays in seeking, reaching, or receiving care) to prioritise important trauma care health system barriers in Karonga, Northern Malawi, and exemplify a holistic health system assessment approach applicable in comparable settings.

Methods and findings

To provide multiple perspectives on each conceptual delay and include data from community-based and facility-based sources, we used 9 methods to examine the injury care health system. The methods were (1) household survey; (2) verbal autopsy analysis; (3) community focus group discussions (FGDs); (4) community photovoice; (5) facility care-pathway process mapping and elucidation of barriers following injury; (6) facility healthcare worker Funding: JW was awarded funding from the Drummond Committee of the Royal Army Medical Corps Charity (https://www.ramcassociation.org. uk/about/the-role-of-the-rhq-ramc), the Royal College of Surgeons of England research fellowship scheme (https://www.rcseng.ac.uk/standards-andresearch/research/fellowships-awards-grants/ fellowships/one-year-surgical-research-fellowship/) and the King's centre for global health partnerships (https://www.kcl.ac.uk/kghp). There are no specific grant numbers associated with these awarded funds. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: CHAM, Christian Health Association of Malawi; FGD, focus group discussion; GIS, geographic information system; GRAMMS, Good Reporting of A Mixed Methods Study; HDSS, Health and Demographic Surveillance Site; LCoGS, Lancet Commission on Global Surgery; LMICs, low- and middle-income countries; MEIRU, Malawi Epidemiology and Intervention Research Unit. survey; (7) facility assessment survey; (8) clinical vignettes for care process quality assessment of facility-based healthcare workers; and (9) geographic information system (GIS) analysis. Empirical data collection took place in Karonga, Northern Malawi, between July 2019 and February 2020. We used a convergent parallel study design concurrently conducting all data collection before subsequently integrating results for interpretation. For each delay, a matrix was created to juxtapose method-specific data relevant to each barrier identified as driving delays to injury care. Using a consensus approach, we graded the evidence from each method as to whether an identified barrier was important within the health system.

We identified 26 barriers to access timely quality injury care evidenced by at least 3 of the 9 study methods. There were 10 barriers at delay 1, 6 at delay 2, and 10 at delay 3. We found that the barriers "cost," "transport," and "physical resources" had the most methods providing strong evidence they were important health system barriers within delays 1 (seek-ing care), 2 (reaching care), and 3 (receiving care), respectively. Facility process mapping provided evidence for the greatest number of barriers—25 of 26 within the integrated analysis. There were some barriers with notable divergent findings between the community- and facility-based methods, as well as among different community- and facility-based methods, which are discussed. The main limitation of our study is that the framework for grading evidence strength for important health system barriers across the 9 studies was done by author-derived consensus; other researchers might have created a different framework.

Conclusions

By integrating 9 different methods, including qualitative, quantitative, community-, patient-, and healthcare worker-derived data sources, we gained a rich insight into the functioning of this health system's ability to provide injury care. This approach allowed more holistic appraisal of this health system's issues by establishing convergence of evidence across the diverse methods used that the barriers of cost, transport, and physical resources were the most important health system barriers driving delays to seeking, reaching, and receiving injury care, respectively. This offers direction and confidence, over and above that derived from single methodology studies, for prioritising barriers to address through health service development and policy.

Author summary

Why was this study done?

- Injuries represent a vast and relatively neglected burden of disease affecting low- and middle-income countries (LMICs).
- While evidence suggests that many health systems underperform in treating injured patients in LMICs, most assessments have not considered all elements of the healthcare system for injured people.
- Innovative mixed methods approaches to holistic health system assessment including community and facility perspectives are therefore needed.

What did the researchers do and find?

- To examine the injury care health system in Karonga, Northern Malawi, we integrated the findings from 9 different methods: (1) household survey; (2) verbal autopsy analysis; (3) community focus group discussions (FGDs); (4) community photovoice; (5) facility care-pathway process mapping and elucidation of barriers following injury; (6) facility healthcare worker surveys; (7) facility assessment surveys; (8) clinical vignettes for care quality assessment of facility-based healthcare workers; and (9) geographic information system (GIS) analysis.
- We graded the strength of evidence each method provided as to whether a given barrier was important in inhibiting access to timely quality injury care.
- We found 26 barriers evidenced by at least 3 of the 9 methods with the barriers "cost," "transport," and "physical resources" having the strongest evidence that they were important barriers delaying seeking, reaching, and receiving care, respectively.

What do these findings mean?

- By comparing the findings from the perspectives of 9 different methods, we were able to gain an in-depth understanding of the health system for trauma care.
- This approach can allows researchers and planners to know the barriers consistently shown to be important and prioritise health service development and policy interventions accordingly.
- This study, to our knowledge, represents a novel and innovative approach in terms of both the number and types of methods mixed, serving as an example other researchers could use in similar contexts.
- The way we graded evidence strength for comparison across methods was somewhat subjective and other researchers may have made different judgements.

Introduction

Injuries resulting from trauma represent a vast and relatively neglected burden of disease, predominantly affecting low- and middle-income countries (LMICs) [1–3]. To ensure injured people can seek, reach, and receive high-quality care when needed requires a whole health system response [4]. However, while studies have shown that health systems underperform in treating injured patients in LMICs [1,5–7], a systematic review of the literature has shown most published assessments have not considered the whole system [8].

Health systems are complex adaptive systems representing a challenging or "wicked" problem to understand [9]. They are built out of historical contexts, rooted within human institutions, influenced by social behaviours governing function and performance, and unpredictably responsive to interventions [10–12]. Meaningful assessments should therefore include the whole health system if the desired outcome is to support impactful health system improvement [13–15].

Systems thinking has encouraged moves away from understanding health systems only through positivist or reductionist approaches [11,16,17]. Constructivist paradigms, which seek

to understand a health system's elements within a social and ecological context, are necessary. Well-designed studies that incorporate mixed method approaches [18], combining the strengths of different underpinning research philosophies, are needed [19]. Such research can allow a better-developed and more nuanced understanding of a problem than is possible through single methodological approaches alone [20].

Many different frameworks exist for describing, understanding, and researching health systems as complex systems with origins in differing paradigms of understanding and sociopolitical backgrounds [21]. A universal framework for understanding complex systems may therefore not exist, and tools selected for a particular assessment should match the intended purpose [21]. The 3 delays framework, which specifically considers factors delaying care seeking (delay 1), reaching a place of care (delay 2), and receiving appropriate and quality care (delay 3) [22], has been widely adopted across reproductive, maternal, newborn, and child health [23–27]. The Lancet Commission on Global Surgery (LCoGS) popularised the framework's use in broader surgical research on barriers in access to quality care [28], and, subsequently, it has become an accepted framework used to evaluate emergency healthcare in LMICs, including trauma [5,29–31].

We used a 3 delays framed mixed method assessment to achieve a well-developed understanding of the whole trauma care health system in Karonga, Northern Malawi. We aimed to integrate findings from 9 diverse methods to show important health system barriers to accessing quality care and thereby exemplify a holistic approach to health system assessment applicable in comparable low-income settings.

Methods

Study setting

Malawi is a landlocked, low-income country; the fourth poorest in the world [32]. Half of its population are children, and 70% of the population lives on less that \$2.15 a day [32,33]. Although limited, available evidence confirms, like in many LMICs, injury is a serious and increasingly important health problem causing substantial and potentially avoidable mortality and morbidity [34]. Most studies on injury care within Malawi have focused on facility-based care [35–37]. There is little evidence available about prehospital care, transport, and reaching care and almost no understanding of population barriers to seeking care following injury, in keeping with the limited global literature on this subject [8].

The study was conducted at the Karonga Health and Demographic Surveillance Site (HDSS) of the Malawi Epidemiology and Intervention Research Unit (MEIRU). It focused specifically on the health system serving the population of the Karonga HDSS in Karonga District, Northern Malawi [38]. The HDSS has a population of over 40,000 (10,228 households) under surveillance since 2002 [38]. It is based in the surrounds of Chilumba, in the south of Karonga, a predominantly rural lakeshore district, typical of a Malawian subsistence economy community [38]. The HDSS population is mostly rural, although about 15% live in semi-urban settlements. Half of the population lives within 1 km of a paved road [38]. The main paved road runs through the district with mostly gravel secondary roads.

The Malawian government provide free facility healthcare to all residents, although out-ofpocket household payments still occur [39]. Traditional healers are recognised as deliverers of health services, although there is little link with the formal health system [40]. The private sector consists of private for-profit and private not-for-profit providers, mainly the Christian Health Association of Malawi (CHAM), which trains healthcare workers and provides services particularly for populations remote from public facilities [41]. The Karonga HDSS population is served by local primary facilities run by the government, including a military facility accessible by civilians, private-for-profit, and CHAM providers. Secondary care facilities include a government facility 70 km north and a CHAM facility 40 km south over difficult hilly terrain. Tertiary care is provided at a government facility in the regional capital, Mzuzu, 150 km south. Most clinicians are non-physicians [42]. Medical Assistants, Clinical Officers, and doctors have completed post-secondary training of a 2-year clinical medicine certificate, 3-year clinical medicine diploma, and 6-year Bachelor's degree, respectively [35]. All provide immediate care to injured patients. Primary facilities are usually staffed by Medical Assistants and Clinical Officers. There is no established prehospital emergency medical service.

Study design

The study design was created following a systematic literature review on trauma health system assessment methods [8], informal literature searches on health system assessment approaches used for other health conditions [43–49], and discussion among coauthors. A range of methods were chosen to holistically assess the health system in its ability to provide injury care. Hence, they cover multiple aspects of care provision and access and multiple perspectives, from community to facility facing. A prior Delphi study [30] to elucidate barriers at each of the 3 delay stages (delay 1—seeking care, delay 2—reaching a place of care, and delay 3—receiving appropriate and quality care) for injury also informed discussions between authors on which methodologies to use to assess those barriers and shaped the design of those methodologies to capture these barriers.

Nine methods to examine the injury care health system were agreed upon which captured all relevant information and were feasible to deliver within MEIRU. These were selected to provide multiple perspectives on each conceptual delay, encompass a range of methodological approaches, and include data from community-based and facility-based sources. The study methods were (1) household survey; (2) verbal autopsy analysis; (3) community focus group discussions (FGDs); (4) community photovoice; (5) facility care-pathway process mapping and elucidation of barriers following injury; (6) facility healthcare worker survey; (7) facility assessment survey; (8) clinical vignettes for care process quality assessment of facility-based healthcare workers; and (9) geographic information system (GIS) analysis. A more detailed description of unpublished study methods is reported in S1 Text.

Study methods involving discussion or interview with educated healthcare workers were conducted in English by JW. Study methods based in the community or involving survey or discussion with less educated healthcare workers (for example, security staff) were conducted in the vernacular language (Chitumbuka). A summary of the study methods included in the mixed methods assessment is in <u>Table 1</u>. Data collection took place between July 2019 and February 2020.

Mixed methods analysis

We used a mixed methods approach and followed the Good Reporting of A Mixed Methods Study (GRAMMS) principles for reporting mixed methods research [50]. We adopted a pragmatic philosophical approach to mixing methods, using a convergent parallel study design concurrently conducting all data collection before subsequently integrating results for interpretation [51,52]. We adopted this approach to allow interpretation of the convergence, divergence, silence, or other relationships of the findings [52–55].

Many study methods were constructed and analysed with the 3 delays framework specifically in mind using a framework of barriers to injury care derived from a previously published Delphi study [30]. This Delphi study developed expert consensus on the most important barriers to consider when evaluating an LMIC health system's preparedness to deliver injury care

Method Community or facility based		Purpose	Sample size achieved	Delay assessing	
Household survey	Community	Nonfatal injury burden description and assessment of health system utilisation.	1,819 households participated.	1, 2, 3	
Verbal autopsy	Community	Assessment of avoidability of fatal injuries within the population and barriers associated with delays to care.	185 injury deaths	1, 2, 3	
FGD	Community	Community perspectives on the barriers driving delays to injury care.	3 FGDs of 7 participants each	1, 2, 3	
Community photovoice	Community	Community perspectives on the barriers driving delays to injury care.	1 group of 7 participants	1, 2, 3	
Facility process mapping [83]	Facility	Healthcare worker descriptions of the process of seeking, reaching and receiving care following injury, and any barriers driving delays.	11 facility maps (54 healthcare workers in total).	1, 2, 3	
Healthcare worker survey	Facility	Healthcare worker perspectives on the barriers driving delays to seeking reaching and receiving care following injury.	228 healthcare workers surveyed	1, 2, 3	
Facility assessment survey			11 facility assessment surveys	3	
Clinical vignettes [96]	Facility	Assessment of the quality of care process provided to injured patients.	85 clinicians	3	
GIS analysis [<u>93</u>]	Not applicable	Description of proximity and physical accessibility of health facilities to the population.	Not Applicable	2	

Table 1. Summary of individual health system assessment study methods used. The details of focus, purpose, sample size, and which delay they assessed are included. The table aims to summarise the methods with more detail available in either the associated publication referenced, or S1 Text.

FGD, focus group discussion; GIS, geographic information system.

https://doi.org/10.1371/journal.pmed.1004344.t001

[30]. Additional barriers generated inductively from at least 3 or more individual study methods were also included. For each delay, a matrix was created to juxtapose the study-specific data relevant to each barrier (Table 2). To assess these data for convergence (the extent to which findings complement or reinforce each other [20]), a consensus exercise was undertaken between 3 authors (JW, IE, and AD). These authors discussed a framework for how barriers identified by each method would be graded according to the evidence provided that they were an important health system barrier. For this purpose, importance of the identified health system barrier was considered a function of both (a) numbers of people potentially affected and (b) the amount of harm potentially caused. Using this framework agreed by these 3 authors, each barrier was graded as having either strong, moderate, or weak evidence that it was important within the health system. Barriers for which a study methodology provided no evidence were categorised as silent. This silence could be a limitation of a method being unsuitable to detect evidence for a barrier, or suggestive that the barrier does not exist. This framework for grading the strength of evidence from each method that a barrier is important is detailed in Table 3. This approach to integrating the study results involved "quantitising" data. Quanititising is a common approach for enabling data transformation and comparison and can be defined as assigning a numerical value to data conceived as nonnumerical [56]. Assigning a descriptive strength grading to quantitative data in this analysis can also be considered a form of qualitising, converting quantitative data into qualitative data [56]. Qualitative

Table 2. Example of the matrix for the mixed method analytical approach. Illustrating the approach to devising a framework for integrating the evidence across the	e
methods.	

Barrier	Study Method 1	Study Method 2	Study Method 3	Etc.
A	Description of findings	Description of findings	Description of findings	
	Grade evidence as an important health system	Grade evidence as an important health system	Grade evidence as an important health system	
	barrier (strong, moderate, weak, silent).	barrier (strong, moderate, weak, silent).	barrier (strong, moderate, weak, silent).	
B etc.				

https://doi.org/10.1371/journal.pmed.1004344.t002

Table 3. Framework for grading strength of evidence in the mixed method cross-study analysis. Qualitative narratives were also included within the matrix for com-
munity FGDs and photovoice. This allowed upgrading of the evidence grade when a narrative strongly suggested importance. Case-by-case decisions were taken where
data were judged to not allow for a standardised approach for all barriers. Details of these decisions are evidenced in the matrix analysis results (S1 Table). Overall delay evi-
dence gradings were similarly case by case.

Study method	Delays and barriers evidenced	Strong Grade	Moderate Grade	Weak Grade	Silent Grade
Household survey	Overall delays 1 to 3, barriers only delay 1.	>10% of injured patients not seeking care report this barrier			Barrier not described
Verbal autopsy	Overall delays 1 to 3, barriers delays 1 to 3.	>20% of avoidable deaths within the delay evidence this barrier	10% to 20% of avoidable deaths within the delay evidence this barrier	<10% of avoidable deaths within the delay evidence this barrier	Barrier not described
Community FGD	Overall delays 1 to 3, barriers delays 1 to 3.	All 3 FGDs describe this barrier	2/3 FGDs describe this barrier.	1/3 FGDs describe this barrier.	Barrier not described
Community photovoice	Overall delays 1 to 3, barriers delays 1 to 3.	3 or more photos describe this barrier	2 photos describe this barrier.	1 photo describes this barrier.	Barrier not described
Facility process mapping	Overall delays 1 to 3, barriers delays 1 to 3.	Frequently reported, i.e., 6–11 facility maps evidence this barrier	Sometimes reported, i.e., 3–5 facility maps evidence this barrier	Rarely reported, i.e., 1–2 facility maps evidence this barrier	Barrier not described
Healthcare worker survey	Overall delays 1 to 3, barriers delays 1 to 3.	Either (a) within first 1/3 most important according to barrier score per delay or (b) within first 1/3 in top 3 overall barrier ranking (grade whichever is highest)	Either (a) within middle 1/3 most important according to barrier score per delay or (b) within middle 1/3 in top 3 overall barrier ranking (grade whichever is highest)	Either (a) within 1/3 least important according to barrier score per delay or (b) within last 1/3 in top 3 overall barrier ranking (grade whichever is highest)	Barrier not described
Facility assessment	Overall delay 3, some barriers in delay 3.	Case by case	Case by case	Case by case	Case by case
Clinical vignettes	Overall delay 3, some barriers in delay 3.	Case by case	Case by case	Case by case	Case by case
GIS analysis	Overall delay 2.				

FGD, focus group discussion; GIS, geographic information system.

https://doi.org/10.1371/journal.pmed.1004344.t003

narratives were also included within the grading framework. This allowed upgrading of the evidence grade when a narrative strongly suggested importance. This grading approach was also taken for the overall evidence for each delay. The facility assessment and clinical vignettes evidence strength grading was judged on a case-by-case basis (by JW and IE) since the data from these methodologies did not fit the standardised approach used for other barriers. As common with convergent parallel approaches, no greater weighting was given to one study method over another [52].

Two authors (JW and IE) subsequently independently tested the agreed scoring framework against 40% sample of barriers (each of the 3 delays overall summaries and 8 individual barriers) exposed by the study methods. A weighted Cohen's kappa test for agreement was performed between the 2 authors' gradings [57]. All areas of disagreement from this sample were discussed until agreement, and a common understanding was reached. One author (JW) then continued to grade the remaining study evidence within the matrix.

Ethical considerations

All participants gave fully informed consent to participate. The Household Survey was approved by the Malawi National Health Sciences Research Committee (ref 19/07/2368) and

the UK MOD Research and Ethics Committee (ref 961/MODEC/19). The remainder of the methods were approved by the Malawi National Health Sciences Research Committee (ref 19/03/2263) and the UK MOD Research and Ethics Committee (ref 960/MODEC/19). Where identifiable images are used, participant consent was obtained, including those from the community photovoice method.

Patient and participant involvement

For the community methods involving primary data collection (household survey, FGDs, and photovoice), a community sensitization meeting took place. Traditional community heads were invited to attend a meeting where all aspects of the community methods were explained, and questions answered allowing onward dissemination to the community. This is routine practice on the introduction of new studies within the MEIRU Karonga HDSS [38].

Results

We identified 26 barriers to access timely quality injury care evidenced by at least 3 of the 9 study methods. There were 10 barriers at delay 1, 6 at delay 2, and 10 at delay 3, of which 20 were also found in the Delphi-generated framework. Cohen's kappa for agreement between authors on the grading of the strength of evidence for importance of barriers was 0.688 (95% confidence interval 0.525, 0.851), demonstrating good agreement [57].

The barriers, their description, and the corresponding strength of evidence that they are important within the studied health system are displayed in matrix form in summary in Table 4 and in detail in S1 Table. Within the table, matrix barriers are displayed in order according to the number of methods finding strong evidence that they are important. Those barriers with the greater number of methods finding strong evidence that they were important are at the top. Within delay 1, the barrier "cost" is at the top with 5 methods finding strong and 1 method moderate evidence that it is an important health system barrier. Within delay 2, the barrier "transport" is at the top with 5 methods (all those applicable) finding strong evidence that it is an important health system barrier "physical resources" is at the top with 7 methods (all those applicable) finding strong evidence that it is an important health system barrier.

Overall, 6 barriers within delay 1 had strong evidence that they were important in the majority of applicable methodologies. These were "cost" (5/6), "healthcare literacy" (4/6), "cultural norms" (4/6), "traditional healers" (3/6), "perceived physical access" (3/6), and "community or bystander engagement" (3/6). Within delay 2, 3 barriers had strong evidence that they were important in the majority of applicable methods, "*transport*" (5/5), "*roads*" (4/5), and "*distance*" (4/5). For delay 3, 5 barriers had strong evidence of being important in the majority of applicable methods, "*transport*" (5/7), "*interfacility transfer*" (5/7), and "*quality processes*" (4/7).

Facility process mapping was the method that provided evidence for the greatest number of barriers, 25 out of 26 within the integrated analysis. This was followed by the healthcare worker survey (24 out of 26), community FGDs (22 out of 26), and verbal autopsy (18 out of 26).

There were some barriers with notable divergent findings between the community- and facility-based methods. This included the barrier "*cultural norms*," which had only weak evidence of importance from community qualitative methods, only being reported in one FGD and not at all in the photovoice evidence, despite strong facility-based evidence. "*Community or bystander engagement*" had only weak evidence of importance from facility process mapping, featuring in 1/11 facility process maps and not raised in the healthcare worker survey, despite strong community-derived evidence. "*Alleged corruption*" was strongly evidenced in

Table 4. Mixed method matrix summary of evidence strength grades according to barrier within each of the 3 delays. Barriers are in order of evidence strength. Not all study methods could, by design, evidence all the delays (labelled as Not Applicable). Barriers for which a study methodology provided no evidence were categorised as silent. This silence could be a limitation of a method being unsuitable to detect evidence for a barrier, or suggestive that the barrier does not exist. A traffic light colour code, green for strong, amber for moderate, and red for weak has been used to facilitate interpretation.

	Household Survey Evidence	Verbal Autopsy Evidence	Community FGD Evidence	Community Photovoice Evidence	Facility Process Mapping Evidence	Healthcare Worker Survey Evidence	GIS Evidence	Facility Assessment Evidence	Clinical Vignettes Evidence
Delay 1 Overall	Moderate	Strong	Strong	Strong	Strong	Strong	Not Applicable	Not Applicable	Not Applicable
Delay 1 Barriers									
COST—The financial costs associated with seeking care are too great	Strong	Moderate	Strong	Strong	Strong	Strong	Not Applicable	Not Applicable	Not Applicable
HEALTHCARE LITERACY- People do not understand about health and available healthcare	Strong	Strong	Strong	Silent	Strong	Moderate	Not Applicable	Not Applicable	Not Applicable
CULTURAL NORMS—Normal cultural behaviours delay seeking care such as gender roles, family responsibilities and requiring someone else's permission to seek care	Strong	Strong	Weak	Silent	Strong	Strong	Not Applicable	Not Applicable	Not Applicable
TRADITIONAL HEALERS- People prefer traditional healers	Moderate	Weak	Strong	Strong	Strong	Moderate	Not Applicable	Not Applicable	Not Applicable
PERCEIVED PHYSICAL ACCESS—People perceive that care is too difficult to physically access	Strong	Silent	Strong	Silent	Strong	Moderate	Not Applicable	Not Applicable	Not Applicable
COMMUNITY OR BYSTANDER ENGAGEMENT-Not enough is done by fellow citizens to support care-seeking	Strong	Silent	Strong	Strong	Weak	Silent	Not Applicable	Not Applicable	Not Applicable
PERCEIVED CARE QUALITY —People perceive that available facility care is poor quality	Moderate	Silent	Strong	Moderate	Strong	Moderate	Not Applicable	Not Applicable	Not Applicable
DELAYED DISCOVERY— There are delays in discovering injured people, including because of intoxication	Weak	Strong	Weak	Silent	Moderate	Weak	Not Applicable	Not Applicable	Not Applicable
RELIGIOUS OR OTHER BELIEFS-believing that seeking formal healthcare is itself wrong	Weak	Silent	Weak	Moderate	Strong	Silent	Not Applicable	Not Applicable	Not Applicable
FEAR OR LACKING COURAGE-irrational incapacitation or rational concern of consequences such as being accused	Weak	Silent	Moderate	Silent	Moderate	Weak	Not Applicable	Not Applicable	Not Applicable
Delay 2 Overall	Moderate	Strong	Strong	Strong	Strong	Strong	Moderate	Not Applicable	Not Applicable
Delay 2 Barriers									
TRANSPORT—There is a lack of timely affordable emergency transport (formal or informal)	Not Applicable	Strong	Strong	Strong	Strong	Strong	Not Applicable	Not Applicable	Not Applicable
ROADS—There is a lack of reliable uncongested roads with priority for emergency vehicles	Not Applicable	Strong	Strong	Strong	Strong	Weak	Not Applicable	Not Applicable	Not Applicable

(Continued)

Table 4. (Continued)

	Household Survey Evidence	Verbal Autopsy Evidence	Community FGD Evidence	Community Photovoice Evidence	Facility Process Mapping Evidence	Healthcare Worker Survey Evidence	GIS Evidence	Facility Assessment Evidence	Clinical Vignettes Evidence
DISTANCE—There is a large physical distance from place of injury to an appropriate healthcare facility	Not Applicable	Strong	Strong	Silent	Strong	Strong	Not Applicable	Not Applicable	Not Applicable
PREHOSPITAL CARE—There is a lack of timely available prehospital emergency care (formal or informal/bystander)	Not Applicable	Strong	Silent	Silent	Weak	Moderate	Not Applicable	Not Applicable	Not Applicable
COMMUNICATION—There is a lack of accessible emergency assistance communication mechanism (e.g., emergency call centre)	Not Applicable	Strong	Silent	Silent	Weak	Moderate	Not Applicable	Not Applicable	Not Applicable
COORDINATION—There is a lack of emergency care service coordination, including bypassing unsuitable facilities	Not Applicable	Strong	Silent	Silent	Weak	Weak	Not Applicable	Not Applicable	Not Applicable
Delay 3 Overall	Weak	Strong	Strong	Strong	Strong	Moderate		Strong	Strong
Delay 3 Barriers									
PHYSICAL RESOURCES— There is a lack of reliably available necessary physical resources (e.g., infrastructure, equipment, and consumable material)	Not Applicable	Strong	Strong	Strong	Strong	Strong	Not Applicable	Strong	Strong
STAFF—In regard to staffing, there is a lack of reliably available, suitably trained, and motivated clinical staff	Not Applicable	Strong	Strong	Moderate	Strong	Strong	Not Applicable	Strong	Strong
SPECIALISTS—There is a lack of reliable, timely access to specialist injury care services	Not Applicable	Strong	Strong	Moderate	Weak	Strong	Not Applicable	Strong	Strong
INTERFACILITY TRANSFER —Lack of available means to safely and quickly transfer injured patients on to a more specialist facility.	Not Applicable	Strong	Strong	Strong	Strong	Strong	Not Applicable	Moderate	Silent
QUALITY PROCESSES—There is a lack of good quality, consistent, structured, clinical priority-driven injury care processes	Not Applicable	Strong	Strong	Moderate	Moderate	Moderate	Not Applicable	Strong	Strong
ALLEGED CORRUPTION— Need for unauthorised payments or gifts to healthcare staff to receive best available treatment (e.g., corruption)	Not Applicable	Silent	Strong	Strong	Silent	Weak	Not Applicable	Silent	Silent
POLICE PROCESSES– perceived or actual police functions affect care access	Not Applicable	Weak	Moderate	Silent	Strong	Weak	Not Applicable	Silent	Silent
PAYMENT—Difficulties with timely payment for care	Not Applicable	Weak	Moderate	Moderate	Weak	Weak	Not Applicable	Silent	Silent

(Continued)

Table 4. (Continued)

	Household Survey Evidence	Verbal Autopsy Evidence	Community FGD Evidence	Community Photovoice Evidence	Facility Process Mapping Evidence	Healthcare Worker Survey Evidence	GIS Evidence	Facility Assessment Evidence	Clinical Vignettes Evidence
PATIENT COOPERATION— There is a lack of patient and family cooperation with care processes	Not Applicable	Silent	Moderate	Silent	Weak	Moderate	Not Applicable	Silent	Silent
CAPACITY—In regard to patient demand, there is insufficient facility capacity to meet patient demand (e.g., overcrowding)	Not Applicable	Silent	Silent	Silent	Moderate	Moderate	Not Applicable	Silent	Silent

https://doi.org/10.1371/journal.pmed.1004344.t004

community qualitative methods (FGDs and photovoice) while only weakly evidenced in the healthcare worker survey and no evidence from facility process mapping, facility assessment, or clinical vignettes.

There was some divergence of findings among the facility-based methods. The barriers "*roads*" and "*police processes*" were strongly evidenced as important in facility process mapping but only weakly in the healthcare worker survey. "*Religious or other beliefs*" were also strongly evidenced as important through facility process mapping but not at all in the healthcare worker survey. Conversely, the barrier "*specialists*" had strong evidence of importance from the healthcare worker survey but was not prioritised in facility process mapping.

Among community-based methods, there was also some observed divergence. The delay 1 barriers "*healthcare literacy*" and "*perceived physical access*" were both strongly evidenced in the household survey and FGDs but not evidenced by the photovoice method. Similarly, the barriers "*distance*," "*police processes*," and "*patient cooperation*" were strongly or moderately evidenced as important by FGDs but not evidenced in the photovoice method.

There was other notable divergence between some of the study methods in evidencing barriers. The delay 2 (reaching care) barriers, "*prehospital care*," "*communication*," and "*coordination*," were strongly evidenced by the verbal autopsy method. Conversely, there was no evidence from the qualitative community methods, and only weak or moderate evidence from the facility-based methods. This contrasted with delay 1 (seeking care), where several barriers including "*perceived physical access*," "*perceived care quality*," "*community or bystander engagement*," and "*religious or other beliefs*" had strong evidence of importance from other methods that were not evidenced by the verbal autopsy method. The same was found for the delay 3 (receiving care) barrier "*alleged corruption*."

Discussion

Our study used multiple methods to holistically capture important barriers to seeking, reaching, and receiving timely quality care following injury. Using a, to the best of our knowledge, novel consensus approach, we graded the evidence from each method as to whether an identified barrier was important within the health system. This grading facilitated integration of the results from individual methods to reveal those barriers with the most methods providing strong evidence that they were an important health system barrier. We used a tabulated matrix to order identified barriers according to the number of methods providing strong evidence they are an important health system barrier. We found that the barriers "cost," "transport," and "physical resources" had the most methods providing strong evidence they were important health system barriers within delays 1, 2, and 3, respectively, suggesting the importance of prioritising these barriers in order to improve the health system's ability to care for people who have been injured.

Integrating findings from 9 diverse methods allowed a deeper understanding of the complexity of the health system for trauma care, viewed from multiple different lenses. It represents a valuable approach in this field where most previous studies assessing barriers to access to care following injury have focused on a single aspect of the health system [8,58]. Most approaches to assessing barriers have also been single method approaches, which can fail to capture both community and facility evidence concurrently [8]. Other attempts to combine methods to understand barriers to injury care have shown, as we have, that using multiple methods identifies important barriers that would have been missed through using single methods alone [59]. Attempts to prioritise the importance of injury care barriers in health systems have been done using Delphi methods [30] and workshop ranking exercises [4,59,60]. However, we used an integrative analytical approach combining qualitising and quantitising techniques to present findings through a joint matrix display, bringing the various data together through a visual format. This process highlighted and ordered barriers with the most methods providing strong evidence they are important for application to health system strengthening, thereby providing insights beyond a siloed analysis of results [52,61]. We found the use of Strong, Moderate, Weak, and Silent data labels in our comparative matrix valuable to supporting integration in our study and propose this as an exemplar for wider application in mixed method health system research [62,63]. Convergent study design joint displays are often themes-by-statistics or side-by-side comparisons; however, due to the unusually large number of different methods being compared in our study, we believe we have showed novel depth to the joint display integration [62] to support quicker interpretation of results [64].

Noteworthy divergent findings within mixed method analysis should be welcomed. They can provide opportunity for creative insight into the methods used or the study context, stimulate interpretive explanations, and set future research priorities [20]. They can also shape choice of future methods for similar assessments to ensure all barriers are evaluated. Divergence of findings between community- and facility-based methods in our study included the barriers "*cultural norms*" with weak evidence from community qualitative methods in contrast to strong facility-based evidence. Community social and cultural perceptions and gender-based factors are known barriers to women seeking care [4,22,30,65,66]. Male household heads can have higher disease severity thresholds for financing health-seeking, more readily financing healthcare for male household members [67]. All of our FGD groups were mixed sex. Such issues might be more likely to arise in single-sex discussion groups [68] as a sensitive topic that only community leaders felt empowered to raise. Community photovoice and FGD participants were not provided with the Delphi [30] derived barriers but generated ideas themselves without opportunity to confirm (or deny) this phenomenon specifically. Future qualitative enquiry could perhaps better explore this issue directly.

"Alleged corruption" was strongly evidenced in community qualitative methods (FGDs and photovoice), while only weakly evidenced in the healthcare worker survey, with no evidence from facility process mapping, facility assessment, or clinical vignettes. Differences between community and facility staff alleging client differentiation based on personal connections or money are common in underresourced health systems [69]. Insights from healthcare workers on this subject may be better obtained through specifically directed discussions [70] rather than the more open approaches we and others have used [69]. Since such practices may have legal or disciplinary implications, a reluctance to discuss them from healthcare workers is understandable [71]. The use of qualitative methods directed to understand this complex and sensitive subject is potentially more suitable and may suspend preconceptions about the controversial subject [71].

Divergence of findings was also seen between facility-based methods. The barrier "*religious* or other beliefs" and "police processes" were strongly evidenced in facility process mapping but only weakly or not at all in the healthcare worker survey. The healthcare worker survey asked directly about a priori Delphi study derived conceptual barriers [30] and offered free text response for participants to propose new ideas. Survey free text responses, commonly used to "safety net" against missing themes [72], were rarely novel and provided by only a minority of participants, as is common for usually unclear reasons [73,74]. Process mapping is interactive and engaging [75] supporting idea generation, particularly from those who understand systems, but infrequently contribute to system-level research and innovation [75]. It also helps "brainstorming" to recall all important aspects, or barriers, in a process [76]. Deep participant engagement for new insights may be less obtainable through surveys than the exercise of process mapping.

The delay 2 barriers "*prehospital care*," "*communication*," and "*coordination*" were strongly evidenced as important by the verbal autopsy method. There was conversely no evidence from the qualitative community methods and only weak or moderate evidence of importance from the facility-based methods. This could reflect the local context of nonexistent formal prehospital emergency care [77,78], unlike other sub-Saharan African contexts where such systems are rapidly developing [79]. Participants cannot want something they do not have prior knowledge of and are therefore unlikely to bring up such issues for discussion without specific prompting. The verbal autopsy analysis was conducted by researchers with an awareness of the potential benefits of formal mature prehospital care systems, who brought this knowledge to the analysis of potential avoidable deaths [5].

Reflecting on our chosen methods, we found process mapping to be highly practical, easy to use, and quickly taught and performed [80], succinctly framing complex health systems problems [81–83]. It evidenced the largest numbers of barriers within our analysis, perhaps as it encouraged participants to consider specific differences between injury mechanisms and contexts. As with any group-based method, workshop discussions may be tempered by hierarchical power dynamics [84,85], perhaps leading to disproportionate contributions from clinicians, especially more senior clinicians, compared to nonclinical staff. Since we used process mapping in facilities with healthcare workers as participants, patient or community perspectives were missing. However, incorporating community members would have required local language facilitation with pragmatic research conduct implications.

The community qualitative, household survey, and verbal autopsy methods provided empirical population-derived data for the integrated analysis. Verbal autopsy data were available because of an established longitudinal surveillance programme in this population [38,86]. While verbal autopsy is a valuable potential source of community-derived insight, its broader application to other populations without existing demographic surveillance is limited. Our household survey allowed access to quantitative community data about nonfatal injuries, including those not presenting for healthcare [87]. It was, however, the most expensive method to undertake, approximating to the combined costs of the other methods. This was due to the large resource requirements to administer in-person surveys in a context where internet or postal surveys are impractical. Community qualitative studies, such as FGDs, can potentially generate insight into health system barriers in a short time frame and be readily replicated in other contexts [88,89]. While we incorporated a photovoice extension to one focus group, the additional time, resources, and ethical considerations need to be balanced with the benefits of potential additional insight and further value of pictural representation of barriers of interest [90].

Considering evaluating delay 2, the availability of powerful computing, along with large-scale and open-source data, has led to growth in GIS methods in health system research [48].

GIS has many attractive features. Various open and freely accessible data sources can be used to provide a theoretical spatial access dimension to the subject of interest [91,92]. However, attempts at validation with patient data have shown they can provide overly optimistic estimates of patient reported time to reach urgent care facilities. Therefore, care must be taken with interpretation of GIS models, which may need to be appropriately adjusted to better reflect population geospatial care access [91,93–96]. Furthermore, access to care is a multiface-ted concept including additional dimensions such as cost, cultural norms, and health system responsiveness [4,5,97]. While the temporal dimension to care access that GIS helps estimate is a fundamental component of care access [98], assessing this together with all the other factors influencing emergency care access necessitates integration with other methods. Our mixed method approach has illustrated how this can be done.

Surveys for structures of care are common across LMIC health systems research. We found, as have others, that facility assessments can be quick to complete [99]. They can be repeated and allow comparison across settings but are not suited to assess either delay 1 or 2 or process and outcome dimensions of care [100]. Outcome measures of care can require longitudinal data, such as verbal autopsy in our study, or trauma registries more commonly. Trauma registries remain limited in their uptake in LMICs and only present in a few locations in Malawi [36,37,101,102]. To complement the facility assessment, we used clinical vignettes to assess provider care quality [45,46,96]. Vignettes are frugal relative to care observation and case note review [103], both being constrained by feasibility, sample size, and reporting accuracy [104]. However, vignettes can facilitate standardised comparison of clinicians across facilities [103] and offer a pragmatic method for assessing provider care quality. Vignettes are low cost compared to alternatives such as observation [105]. They also do not depend on either the providers' routine workload nor on patient selection [106,107], and we found vignettes could be undertaken fairly quickly, around 30 minutes per clinician.

In the relatively underresourced and underresearched field of trauma in LMICs [108], efficiency in health system assessments is needed to identify needs and monitor interventions [109,110]. Rapid assessment principles appraise the potential value of methods' suitability for inclusion in efficient and holistic mixed-method health system assessment [111]. These principles are cost-effectiveness, speed, use of multiple data sources, and pragmatism [111]. Cost and speed are intersecting principles since additional resources may speed up assessment processes. Pragmatic metrics are achievable, not arduous to collect, readily interpretable, and widely applicable [112], providing sufficient rather than perfect information [111]. Triangulating findings from multiple data sources to evaluate convergence or divergence of findings is an inherent feature of mixed methods research [52].

Summarised in <u>Table 5</u>, with more detailed explanation in <u>S2 Table</u>, we include our reflections on how the methods in this study perform in line with these principles and their potential for adaptation for future use to improve compliance with these principles of efficient health system assessment. Our reflections are summarised in a grading of well aligned, partly aligned, or poorly aligned.

According to these reflections, process mapping was the most compliant method, scoring as well-aligned with all 4 principles for both performance and potential. We would strongly advocate for its wider adoption by others for including in mixed-method health system assessment. Community FGDs similarly performed well, and we would recommend to include this in assessments also. Pragmatic and cost considerations around administration resource requirements constrain our recommendations for inclusion of both household and health care worker surveys. Similarly, challenges in triangulating and integrating the findings from GIS, verbal autopsy, and clinical vignettes should be carefully considered before including in future mixed-method health system assessments. That being said, the specific purpose for a health

Method		Speed	Pragmatism	Cost-effectiveness	Triangulation
Household survey	Performance	Partly well-aligned	Partly well-aligned	Poorly aligned	Partly well-aligned
	Potential	Well-aligned	Partly well-aligned	Poorly aligned	Well-aligned
Verbal autopsy	Performance	Well-aligned	Well-aligned	Well-aligned	Poorly aligned
	Potential	Well-aligned	Poorly aligned	Well-aligned	Poorly aligned
FGDs	Performance	Partly well-aligned	Well-aligned	Well-aligned	Well-aligned
	Potential	Well-aligned	Well-aligned	Well-aligned	Well-aligned
Community photovoice	Performance	Well-aligned	Partly well-aligned	Well-aligned	Partly well-aligned
	Potential	Well-aligned	Well-aligned	Well-aligned	Well-aligned
Process mapping	Performance	Well-aligned	Well-aligned	Well-aligned	Well-aligned
	Potential	Well-aligned	Well-aligned	Well-aligned	Well-aligned
Healthcare worker survey	Performance	Poorly aligned	Partly well-aligned	Partly well-aligned	Well-aligned
	Potential	Well-aligned	Partly well-aligned	Well-aligned	Well-aligned
GIS analysis	Performance	Well-aligned	Well-aligned	Well-aligned	Partly well-aligned
	Potential	Well-aligned	Well-aligned	Well-aligned	Partly well-aligned
Clinical vignettes	Performance	Partly well-aligned	Well-aligned	Well-aligned	Partly well-aligned
	Potential	Well-aligned	Well-aligned	Well-aligned	Partly well-aligned
Facility assessment	Performance	Well-aligned	Well-aligned	Well-aligned	Partly well-aligned
	Potential	Well-aligned	Well-aligned	Well-aligned	Well-aligned

Table 5. Summary of reflections on the compliance of study methods with the rapid assessment principles (authors' judgement). Performance refers to authors' experience, and potential refers to the scope for adaptation to improve compliance with a rapid assessment principle. These represent the authors' reflections. Well-aligned (green), partly well-aligned (yellow), or poorly aligned (red).

FGD, focus group discussion; GIS, geographic information system.

https://doi.org/10.1371/journal.pmed.1004344.t005

system assessment should nuance the choice of methods included. For example, including photovoice, while perhaps not adding substantially to information about barriers to care in this study, could provide images that may be useful for advocating for intervention [113]. Similarly, if methods, such as process mapping, are planned to take place within a facility, including a facility assessment may be justifiable at the same visit without adding substantially to the cost or time required for the study.

We propose, therefore, that the choice of methods used for future holistic health system assessment should balance (a) the pragmatism of encompassing suitable preexisting data sources, such as verbal autopsy and GIS in our study or trauma registry data where available; (b) the economy of low cost and practical methods such as community focus groups, facility process mapping, and facility assessments; and (c) the resource requirements of methods such as vignettes, or especially administered surveys to either households or healthcare workers. The choices will depend to some degree on specific health system research aims, local research infrastructure and data, and resources available for a given project.

There are limitations to this study. Since we chose a convergent parallel design, a more indepth exploration of some of the observed divergent findings might have been possible through a sequential design. For example, in an explanatory format, targeting FGDs towards results of the surveys, or an exploratory format, designing the survey questions after community qualitative and facility process mapping methods had been analysed. However, sequential approaches would likely be more time consuming, costly, and still have potential to miss important concepts. The integrated matrix analysis focused on the comparison of barriers within each delay, rather than across all delays, which could have been alternatively explored. Our framework for grading evidence strength for each barrier might have had different thresholds if different individuals were constructing it. Similarly, the kappa for agreement could have been stronger. However, our broad approach to devise and undertake a coherent analytical approach to meaningfully compare multiple different study methods remains transferable, and we would encourage other researchers to further explore and test frameworks for meaningful cross-method integration and interpretation. We used an analytical strategy that provided equal weight to each method. Similarly, equal weighting was given to participants in each method. It might have been possible to provide hierarchical bias to responses, for example, in the FGDs, injured participants may have been considered to have more insight into injury care quality, or the community leaders may have been considered more authoritative on issues related to community engagement and behaviour. Likewise, facility process maps were a group-derived output, which, although potentially influenced by personality, cultural and hierarchical roles, were not weighted to value the clinician's views more than those of nonclinicians. As well, this study was limited to a rural population. Urban and rural populations in sub-Saharan Africa experience differing burdens of injury [114]. The scientific value of dichotomising rural and urban contexts has been challenged for some health-related questions [115]. However, the health system function and barriers for accessing high-quality care would still be a valuable frame for such a comparison. Future application of this approach could include urban population comparisons to incorporate any differences. Finally, what was found to be important in Malawi may differ from other contexts, which may, in turn, influence the methodologies required to identify them.

The potential value of mixed methods in emergency health systems research is recognised, but its use is in its infancy [58]. True mixed-methods studies integrating data before drawing conclusions in trauma care system assessments are rare [8]. Our study, therefore, makes an important contribution to the existing literature in the application of mixed methods approaches to health system research in general, and trauma care system research in particular. We are not aware of any similar in-depth assessment on a single population's health system using as many as 9 different methods. Our study, to our knowledge, represents a novel approach in both the weight and focus of a mixed-method study. We found the barriers of cost, transportation, and physical resources were most strongly evidenced across the methods. While previous single method studies have highlighted these health system issues [8], our study has found them to be the most important barriers in the health system that we assessed and should end further speculation on their relative contribution to delays in accessing quality care. We hope this encourages other researchers to use similar approaches to holistically assess health systems in other contexts. Drawing on this study's experience, funding has been granted for further holistic mixed method trauma health system assessment across 4 LMICs [116].

Supporting information

S1 GRAMMS Checklist. Good Reporting of A Mixed Methods Study (GRAMMS checklist). (DOCX)

S1 Text. Unpublished methods supporting information. (DOCX)

S2 Text. Inclusivity in global research questionnaire. (DOCX)

S1 Table. Mixed method integrated analysis detail of evidence grading for each study, for each of the 3 delays and identified barriers. (DOCX) S2 Table. Detail of reflections on the compliance of study methods with the rapid assessment principles (authors' judgement). (DOCX)

Author Contributions

Conceptualization: John Whitaker, Idara Edem, Rory Rickard, Andrew JM Leather, Justine Davies.

Data curation: John Whitaker, Ella Togun.

Formal analysis: John Whitaker, Idara Edem, Ella Togun, Albert Dube, Giulia Brunelli, Thomas Van Boeckel.

Funding acquisition: John Whitaker, Rory Rickard, Andrew JM Leather, Justine Davies.

Investigation: John Whitaker, Ella Togun, Abena S. Amoah, Albert Dube.

Methodology: John Whitaker, Albert Dube, Andrew JM Leather, Justine Davies.

- Project administration: John Whitaker, Ella Togun, Abena S. Amoah, Lindani Chirwa, Boston Munthali.
- Resources: John Whitaker, Abena S. Amoah, Lindani Chirwa, Boston Munthali, Andrew JM Leather, Justine Davies.
- Supervision: Abena S. Amoah, Albert Dube, Rory Rickard, Andrew JM Leather, Justine Davies.

Validation: John Whitaker, Idara Edem, Albert Dube.

Visualization: John Whitaker.

Writing - original draft: John Whitaker.

Writing - review & editing: John Whitaker, Idara Edem, Ella Togun, Abena S. Amoah, Albert Dube, Lindani Chirwa, Boston Munthali, Giulia Brunelli, Thomas Van Boeckel, Rory Rickard, Andrew JM Leather, Justine Davies.

References

- Mock C, Joshipura M, Arreola-Risa C, Quansah R. An estimate of the number of lives that could be 1. saved through improvements in trauma care globally. World J Surg. 2012; 36(5):959-963. Epub 2012/ 03/16. https://doi.org/10.1007/s00268-012-1459-6 PMID: 22419411.
- 2. Meara JG, Greenberg SLM. Global surgery as an equal partner in health: no longer the neglected stepchild. Lancet Glob Health. 2015; 3:S1-S2. https://doi.org/10.1016/S2214-109X(15)70019-7 PMID: 25926313
- Haagsma JA, Graetz N, Bolliger I, Naghavi M, Higashi H, Mullany EC, et al. The global burden of 3. injury: incidence, mortality, disability-adjusted life years and time trends from the Global Burden of Disease study 2013. Inj Prev. 2016; 22(1):3-18. Epub 2015/12/05. https://doi.org/10.1136/injuryprev-2015-041616 PMID: 26635210; PubMed Central PMCID: PMC4752630.
- Odland ML, Whitaker J, Nepogodiev D, Aling' CA, Bagahirwa I, Dushime T, et al. Identifying, Prioritiz-4. ing and Visually Mapping Barriers to Injury Care in Rwanda: A Multi-disciplinary Stakeholder Exercise. World J Surg. 2020; 44(9):2903-2918. https://doi.org/10.1007/s00268-020-05571-6 PMID: 32440950
- 5. Edem IJ, Dare AJ, Byass P, D'Ambruoso L, Kahn K, Leather AJM, et al. External injuries, trauma and avoidable deaths in Agincourt, South Africa: a retrospective observational and qualitative study. BMJ Open. 2019; 9(6):e027576. https://doi.org/10.1136/bmjopen-2018-027576 PMID: 31167869
- Kuzma K, Lim AG, Kepha B, Nalitolela NE, Reynolds TA. The Tanzanian trauma patients' prehospital 6. experience: a qualitative interview-based study. BMJ Open. 2015; 5(4). https://doi.org/10.1136/ bmjopen-2014-006921 PMID: 25916487

- Pouramin P, Li CS, Busse JW, Sprague S, Devereaux PJ, Jagnoor J, et al. Delays in hospital admissions in patients with fractures across 18 low-income and middle-income countries (INORMUS): a prospective observational study. Lancet Glob Health. 2020; 8(5):e711–e720. https://doi.org/10.1016/ S2214-109X(20)30067-X PMID: 32353318
- Whitaker J, Donohoe N, Denning M, Poenaru D, Guadagno E, Leather AJM, et al. Assessing trauma care systems in low-income and middle-income countries: a systematic review and evidence synthesis mapping the Three Delays framework to injury health system assessments. BMJ Glob Health. 2021; 6(5):e004324. https://doi.org/10.1136/bmjgh-2020-004324 PMID: 33975885
- Finegood DT. Can We Build an Evidence Base on the Impact of Systems Thinking for Wicked Problems? Comment on "What Can Policy-Makers Get Out of Systems Thinking? Policy Partners' Experiences of a Systems-Focused Research Collaboration in Preventive Health". Int J Health Policy Manag. 2021; 10(6):351–353. Epub 2020/10/17. https://doi.org/10.34172/ijhpm.2020.194 PMID: 33059430.
- Alliance for Health Policy and Systems Research. Systems thinking for health systems strengthening. 2009. Available from: https://ahpsr.who.int/publications/i/item/2009-11-13-systems-thinking-forhealth-systems-strengthening.
- 11. Adam T. Advancing the application of systems thinking in health. Health Res Policy Syst. 2014; 12 (1):50. https://doi.org/10.1186/1478-4505-12-50 PMID: 25160646
- Gilson L. Trust and the development of health care as a social institution. Soc Sci Med. 2003; 56 (7):1453–1468. Epub 2003/03/05. https://doi.org/10.1016/s0277-9536(02)00142-9 PMID: 12614697.
- Remick KN, Wong EG, Chuot Chep C, Morton RT, Monsour A, Fisher D, et al. Development of a novel Global Trauma System Evaluation Tool and initial results of implementation in the Republic of South Sudan. Injury. 2014; 45(11):1731–1735. Epub 2014/09/07. <u>https://doi.org/10.1016/j.injury.2014.08.</u> 004 PMID: 25192865.
- Kumah E, Ankomah SE, Fusheini A, Sarpong EK, Anyimadu E, Quist A, et al. Frameworks for health systems performance assessment: how comprehensive is Ghana's holistic assessment tool? Glob Health Res Policy. 2020; 5(1):10. https://doi.org/10.1186/s41256-020-00139-2 PMID: 32166129
- Bennett S, Peters DH. Assessing National Health Systems: Why and How. Health Syst Reform. 2015; 1(1):9–17. https://doi.org/10.1080/23288604.2014.997107 PMID: 31519082
- Peters DH. The application of systems thinking in health: why use systems thinking? Health Res. Policy Syst. 2014; 12:51. Epub 2014/08/28. https://doi.org/10.1186/1478-4505-12-51 PMID: 25160707; PubMed Central PMCID: PMC4245196.
- Atun R. Health systems, systems thinking and innovation. Health Policy Plan. 2012; 27(Suppl 4):iv4– iv8. Epub 2012/10/04. https://doi.org/10.1093/heapol/czs088 PMID: 23014152.
- Koome G, Atela M, Thuita F, Egondi T. Health system factors associated with post-trauma mortality at the prehospital care level in Africa: a scoping review. Trauma Surg Acute Care Open. 2020; 5(1): e000530. https://doi.org/10.1136/tsaco-2020-000530 PMID: 33083557.
- Johnson RB, Onwuegbuzie AJ. Mixed Methods Research: A Research Paradigm Whose Time Has Come. Educ Res. 2004; 33(7):14–26. https://doi.org/10.3102/0013189x033007014
- Bazeley P. Integrating Analyses in Mixed Methods Research. 55 City Road, London: SAGE Publications Ltd; 2018.
- van Olmen J, Marchal B, Van Damme W, Kegels G, Hill PS. Health systems frameworks in their political context: framing divergent agendas. BMC Public Health. 2012; 12:774. https://doi.org/10.1186/ 1471-2458-12-774 PMID: 22971107; PubMed Central PMCID: PMC3549286.
- Thaddeus S, Maine D. Too far to walk: maternal mortality in context. Soc Sci Med. 1994; 38(8):1091– 110. Epub 1994/04/01. https://doi.org/10.1016/0277-9536(94)90226-7 PMID: 8042057.
- Combs Thorsen V, Sundby J, Malata A. Piecing together the maternal death puzzle through narratives: the three delays model revisited. PLoS ONE. 2012; 7(12):e52090. Epub 2013/01/04. https://doi. org/10.1371/journal.pone.0052090 PMID: 23284882; PubMed Central PMCID: PMC3526530.
- Wilmot E, Yotebieng M, Norris A, Ngabo F. Missed Opportunities in Neonatal Deaths in Rwanda: Applying the Three Delays Model in a Cross-Sectional Analysis of Neonatal Death. Matern Child Health J. 2017; 21(5):1121–1129. Epub 2017/02/20. <u>https://doi.org/10.1007/s10995-016-2210-y</u> PMID: 28214925.
- 25. Waiswa P, Kallander K, Peterson S, Tomson G, Pariyo GW. Using the three delays model to understand why newborn babies die in eastern Uganda. Trop Med Int Health. 2010; 15(8):964–972. Epub 2010/07/20. https://doi.org/10.1111/j.1365-3156.2010.02557.x PMID: 20636527.
- Upadhyay RP, Rai SK, Krishnan A. Using Three Delays Model to Understand the Social Factors Responsible for Neonatal Deaths in Rural Haryana, India J Trop Pediatr. 2013; 59(2):100–105. https:// doi.org/10.1093/tropej/fms060 PMID: 23174990

- 27. Pajuelo MJ, Anticona Huaynate C, Correa M, Mayta Malpartida H, Ramal Asayag C, Seminario JR, et al. Delays in seeking and receiving health care services for pneumonia in children under five in the Peruvian Amazon: a mixed-methods study on caregivers' perceptions. BMC Health Serv Res. 2018; 18(1):149. Epub 2018/03/02. https://doi.org/10.1186/s12913-018-2950-z PMID: 29490643; PubMed Central PMCID: PMC5831863.
- Meara JG, Leather AJ, Hagander L, Alkire BC, Alonso N, Ameh EA, et al. Global Surgery 2030: evidence and solutions for achieving health, welfare, and economic development. Lancet. 2015; 386 (9993):569–624. Epub 2015/05/01. <u>https://doi.org/10.1016/S0140-6736(15)60160-X</u> PMID: 25924834.
- Calvello EJ, Skog AP, Tenner AG, Wallis LA. Applying the lessons of maternal mortality reduction to global emergency health. Bull World Health Organ. 2015; 93(6):417–423. Epub 2015/08/05. https:// doi.org/10.2471/BLT.14.146571 PMID: 26240463; PubMed Central PMCID: PMC4450708.
- Whitaker J, Nepogodiev D, Leather A, Davies J. Assessing barriers to quality trauma care in low and middle-income countries: A Delphi study. Injury. 2020; 51(2):278–285. Epub 2019/12/31. <u>https://doi.org/10.1016/j.injury.2019.12.035</u> PMID: 31883865.
- **31.** Fraser A, Newberry Le Vay J, Byass P, Tollman S, Kahn K, Ambruoso L, et al. Time-critical conditions: assessment of burden and access to care using verbal autopsy in Agincourt, South Africa. BMJ Glob Health. 2020; 5(4):e002289. https://doi.org/10.1136/bmjgh-2020-002289 PMID: 32377406
- World Bank. Poverty and Equity Brief, Malawi, April 2023 [accessed 2023 Dec 11]. Available from: https://databankfiles.worldbank.org/public/ddpext_download/poverty/987B9C90-CB9F-4D93-AE8C-750588BF00QA/current/Global_POVEQ_MWI.pdf.
- UNICEF. Child Poverty in Malawi, Policy Brief. 2018 [accessed 2023 Dec 11]. Available from: https:// www.unicef.org/esa/media/8451/file/UNICEF-Malawi-2018-Child-Poverty-in-Malawi-Policy-Brief.pdf.
- **34.** Commission MNP. The Malawi Non-Communicable Diseases & Injuries Poverty Commission Report. 2018. Available from: https://gene-bukhman-z43v.squarespace.com/malawi-report.
- Chokotho L, Mulwafu W, Singini I, Njalale Y, Jacobsen KH. Improving hospital-based trauma care for road traffic injuries in Malawi. World J Emerg Med. 2017; 8(2):85–90. Epub 2017/05/02. <u>https://doi.org/10.5847/wjem.j.1920-8642.2017.02.001</u> PMID: <u>28458750</u>; PubMed Central PMCID: PMC5409240.
- Chokotho LC, Mulwafu W, Nyirenda M, Mbomuwa FJ, Pandit HG, Le G, et al. Establishment of trauma registry at Queen Elizabeth Central Hospital (QECH), Blantyre, Malawi and mapping of high risk geographic areas for trauma. World J Emerg Med. 2019; 10(1):33–41. <u>https://doi.org/10.5847/wjem.j.</u> 1920-8642.2019.01.005 PMID: 30598716.
- Banza LN, Gallaher J, Dybvik E, Charles A, Hallan G, Gjertsen J-E, et al. The rise in road traffic injuries in Lilongwe, Malawi: A snapshot of the growing epidemic of trauma in low income countries. Int J Surg Open. 2018; 10:55–60. https://doi.org/10.1016/j.ijso.2017.11.004
- Crampin AC, Dube A, Mboma S, Price A, Chihana M, Jahn A, et al. Profile: the Karonga Health and Demographic Surveillance System. Int J Epidemiol. 2012; 41(3):676–685. Epub 2012/06/21. https:// doi.org/10.1093/ije/dys088 PMID: 22729235.
- Bijlmakers L, Wientjes M, Mwapasa G, Cornelissen D, Borgstein E, Broekhuizen H, et al. Out-ofpocket payments and catastrophic household expenditure to access essential surgery in Malawi—A cross-sectional patient survey. Ann Med Surg. 2019; 43:85–90. https://doi.org/10.1016/j.amsu.2019. 06.003 PMID: 31304010
- Lampiao F, Chisaka J, Clements C. Communication Between Traditional Medical Practitioners and Western Medical Professionals. Front Sociol. 2019; 4(37). https://doi.org/10.3389/fsoc.2019.00037 PMID: 33869360
- Munthali AC, Mannan H, MacLachlan M, Swartz L, Makupe CM, Chilimampunga C. Non-use of Formal Health Services in Malawi: Perceptions from Non-users. Malawi Med J. 2014; 26(4):126–132. PMID: 26167263.
- Mullan F, Frehywot S. Non-physician clinicians in 47 sub-Saharan African countries. Lancet. 2007; 370(9605):2158–2163. https://doi.org/10.1016/S0140-6736(07)60785-5 PMID: 17574662
- Ardrey J, Desmond N, Tolhurst R, Mortimer K. The Cooking and Pneumonia Study (CAPS) in Malawi: A Nested Pilot of Photovoice Participatory Research Methodology. PLoS ONE. 2016; 11(6): e0156500. Epub 2016/06/03. https://doi.org/10.1371/journal.pone.0156500 PMID: 27254291; PubMed Central PMCID: PMC4890783.
- Aaronson E, Mort E, Soghoian S. Mapping the process of emergency care at a teaching hospital in Ghana. Healthc (Amst). 2017; 5(4):214–220. Epub 2017/01/10. https://doi.org/10.1016/j.hjdsi.2016. 12.001 PMID: 28065392.
- 45. Vesel L, Manu A, Lohela TJ, Gabrysch S, Okyere E, Ten Asbroek AH, et al. Quality of newborn care: a health facility assessment in rural Ghana using survey, vignette and surveillance data. BMJ Open.

2013; 3(5). Epub 2013/05/15. https://doi.org/10.1136/bmjopen-2012-002326 PMID: 23667161; PubMed Central PMCID: PMC3651975.

- 46. Villar Uribe M, Alonge OO, Bishai DM, Bennett S. Can task-shifting work at scale?: Comparing clinical knowledge of non-physician clinicians to physicians in Nigeria. BMC Health Serv Res. 2018; 18 (1):308. https://doi.org/10.1186/s12913-018-3133-7 PMID: 29716609
- Chirowodza A, van Rooyen H, Joseph P, Sikotoyi S, Richter L, Coates T. Using participatory methods and geographic information systems (GIS) to prepare for an HIV community-based trial in Vulindlela, South Africa (Project Accept–HPTN 043). J Community Psychol. 2009; 37(1):41–57. https://doi.org/ 10.1002/jcop.20294 PMC2747527. PMID: 19774224
- Tanser FC, le Sueur D. The application of geographical information systems to important public health problems in Africa. Int J Health Geogr. 2002; 1(1):4. https://doi.org/10.1186/1476-072x-1-4 PMID: 12537589
- Makanga PT, Schuurman N, von Dadelszen P, Firoz T. A scoping review of geographic information systems in maternal health. Int J Gynaecol Obstet. 2016; 134(1):13–17. Epub 2016/04/01. https://doi.org/10.1016/j.ijgo.2015.11.022 PMID: 27126906.
- O'cathain A, Murphy E, Nicholl J. The Quality of Mixed Methods Studies in Health Services Research. J Health Serv Res Policy. 2008; 13(2):92–98. https://doi.org/10.1258/jhsrp.2007.007074 PMID: 18416914.
- Jick TD. Mixing Qualitative and Quantitative Methods: Triangulation in Action. Adm Sci Q. 1979; 24 (4):602–611. https://doi.org/10.2307/2392366
- Creswell JW, Clark VLP. Designing and conducting mixed methods research. 3rd ed. Sage Publications; 2017.
- Farmer T, Robinson K, Elliott SJ, Eyles J. Developing and implementing a triangulation protocol for qualitative health research. Qual Health Res. 2006; 16(3):377–394. Epub 2006/02/02. https://doi.org/ 10.1177/1049732305285708 PMID: 16449687.
- 54. von der Lippe H. Motivation and Selection Processes in a Biographical Transition: A Psychological Mixed Methods Study on the Transition Into Fatherhood. J Mixed Methods Res. 2010; 4(3):199–221. https://doi.org/10.1177/1558689810365149
- 55. Fitzpatrick KR. Points of Convergence in Music Education: The Use of Data Labels as a Strategy for Mixed Methods Integration. J Mixed Methods Res. 2014; 10(3):273–291. <u>https://doi.org/10.1177/</u> 1558689814560264
- Nzabonimpa JP. Quantitizing and qualitizing (im-)possibilities in mixed methods research. Method Innov. 2018; 11(2):2059799118789021. https://doi.org/10.1177/2059799118789021
- 57. Altman DG. Practical statistics for medical research. London: Chapman and Hall; 1991 1991/10/01. 611 p.
- Mehmood A, Rowther AA, Kobusingye O, Hyder AA. Assessment of pre-hospital emergency medical services in low-income settings using a health systems approach. Int J Emerg Med. 2018; 11(1):53. https://doi.org/10.1186/s12245-018-0207-6 PMID: 31179939
- 59. Odland ML, Abdul-Latif A-M, Ignatowicz A, Alayande B, Appia Ofori B, Balanikas E, et al. Equitable access to quality trauma systems in low-income and middle-income countries: assessing gaps and developing priorities in Ghana, Rwanda and South Africa. BMJ Glob Health. 2022; 7(4):e008256. https://doi.org/10.1136/bmjgh-2021-008256 PMID: 35410954
- Mac Quene T, Smith L, Odland ML, Levine S, D'Ambruoso L, Davies J, et al. Prioritising and mapping barriers to achieve equitable surgical care in South Africa: a multi-disciplinary stakeholder workshop. Glob Health Action. 2022; 15(1):2067395. <u>https://doi.org/10.1080/16549716.2022.2067395</u> PMID: 35730572
- Fetters MD, Curry LA, Creswell JW. Achieving integration in mixed methods designs-principles and practices. Health Serv Res. 2013; 48(6 Pt 2):2134–2156. Epub 2013/10/23. <u>https://doi.org/10.1111/</u> 1475-6773.12117 PMID: 24279835.
- Guetterman TC, Fetters MD, Creswell JW. Integrating Quantitative and Qualitative Results in Health Science Mixed Methods Research Through Joint Displays. Ann Fam Med. 2015; 13(6):554–561. Epub 2015/11/11. <u>https://doi.org/10.1370/afm.1865</u> PMID: 26553895; PubMed Central PMCID: PMC4639381.
- 63. Shaw EK, Ohman-Strickland PA, Piasecki A, Hudson SV, Ferrante JM, McDaniel RR Jr., et al. Effects of facilitated team meetings and learning collaboratives on colorectal cancer screening rates in primary care practices: a cluster randomized trial. Ann Fam Med. 2013; 11(3):220–228. <u>https://doi.org/10.1370/afm.1505</u> PMID: 23690321.
- **64.** Burgess-Allen J, Owen-Smith V. Using mind mapping techniques for rapid qualitative data analysis in public participation processes. Health Expect. 2010; 13(4):406–415. <u>https://doi.org/10.1111/j.1369-7625.2010.00594.x PMID: 20550595</u>

- Capuno JJ, Kraft AD, Poco LC, Quimbo SA, Tan CAR. Health conditions, payments, proximity, and opportunity costs: Examining delays in seeking inpatient and outpatient care in the Philippines. Soc Sci Med. 2019; 238:112479. https://doi.org/10.1016/j.socscimed.2019.112479 PMID: 31421350
- 66. Tounsi LL, Daebes HL, Gerdin Wärnberg M, Nerlander M, Jaweed M, Mamozai BA, et al. Association Between Gender, Surgery and Mortality for Patients Treated at Médecins Sans Frontières Trauma Centre in Kunduz, Afghanistan. World J Surg. 2019; 43(9):2123–2130. Epub 2019/05/09. https://doi. org/10.1007/s00268-019-05015-w PMID: 31065777.
- Onah MN, Horton S. Male-female differences in households' resource allocation and decision to seek healthcare in south-eastern Nigeria: Results from a mixed methods study. Soc Sci Med. 2018; 204:84–91. Epub 2018/03/31. https://doi.org/10.1016/j.socscimed.2018.03.033 PMID: 29602090.
- **68.** Hollander JA. The Social Contexts of Focus Groups. J Contemp Ethnogr. 2004; 33(5):602–637. https://doi.org/10.1177/0891241604266988
- Willott C, Boyd N, Wurie H, Smalle I, Kamara TB, Davies JI, et al. Staff recognition and its importance for surgical service delivery: a qualitative study in Freetown, Sierra Leone. Health Policy Plan. 2021; 36(1):93–100. https://doi.org/10.1093/heapol/czaa131 PMID: 33246332
- 70. Mæstad O, Mwisongo A. Informal payments and the quality of health care: Mechanisms revealed by Tanzanian health workers. Health Policy. 2011; 99(2):107–115. Epub 2010/08/17. https://doi.org/10. 1016/j.healthpol.2010.07.011 PMID: 20709420.
- Stringhini S, Thomas S, Bidwell P, Mtui T, Mwisongo A. Understanding informal payments in health care: motivation of health workers in Tanzania. Hum Resour Health. 2009; 7:53. Epub 2009/07/02. https://doi.org/10.1186/1478-4491-7-53 PMID: 19566926; PubMed Central PMCID: PMC2711965.
- 72. O'Cathain A, Thomas KJ. "Any other comments?" Open questions on questionnaires—a bane or a bonus to research? BMC Med Res Methodol. 2004; 4(1):25. https://doi.org/10.1186/1471-2288-4-25 PMID: 15533249
- Open-ended questions in surveys of patients' satisfaction with family doctors. J Health Serv Res Policy. 2007; 12(2):86–89. https://doi.org/10.1258/135581907780279639 PMID: 17407657
- 74. Garcia J, Evans J, Reshaw M. "Is There Anything Else You Would Like to Tell Us"—Methodological Issues in the Use of Free-Text Comments from Postal Surveys. Quality Quantity. 2004; 38(2):113–125. https://doi.org/10.1023/B:QUQU.0000019394.78970.df
- Trebble TM, Hansi N, Hydes T, Smith MA, Baker M. Process mapping the patient journey: an introduction. BMJ. 2010; 341:c4078. Epub 2010/08/17. https://doi.org/10.1136/bmj.c4078 PMID: 20709715.
- Heher YK, Chen Y. Process mapping: A cornerstone of quality improvement. Cancer Cytopathol. 2017; 125(12):887–890. Epub 2017/11/23. https://doi.org/10.1002/cncy.21946 PMID: 29165909.
- 77. Chokotho L, Mulwafu W, Singini I, Njalale Y, Maliwichi-Senganimalunje L, Jacobsen KH. First Responders and Prehospital Care for Road Traffic Injuries in Malawi. Prehosp Disaster Med. 2017; 32 (1):14–19. Epub 2016/12/07. https://doi.org/10.1017/S1049023X16001175 PMID: 27923422
- Agarwal-Harding KJ, Chokotho LC, Mkandawire NC, Martin C Jr., Losina E, Katz JN. Risk Factors for Delayed Presentation Among Patients with Musculoskeletal Injuries in Malawi. J Bone Joint Surg Am. 2019; 101(10):920–931. Epub 2019/05/17. https://doi.org/10.2106/JBJS.18.00516 PMID: 31094984; PubMed Central PMCID: PMC6530973.
- 79. Jayaraman S, Ntirenganya F, Nkeshimana M, Rosenberg A, Dushime T, Kabagema I, et al. Building Trauma and EMS Systems Capacity in Rwanda: Lessons and Recommendations. Ann Glob Health. 2021; 87(1):104. Epub 2021/11/11. https://doi.org/10.5334/aogh.3324 PMID: 34754760; PubMed Central PMCID: PMC8555476.
- Antonacci G, Reed JE, Lennox L, Barlow J. The use of process mapping in healthcare quality improvement projects. Health Serv Manage Res. 2018; 31(2):74–84. Epub 2018/05/01. <u>https://doi.org/10.1177/0951484818770411</u> PMID: 29707978.
- Durski KN, Naidoo D, Singaravelu S, Shah AA, Djingarey MH, Formenty P, et al. Systems thinking for health emergencies: use of process mapping during outbreak response. BMJ Glob Health. 2020; 5 (10):e003901. https://doi.org/10.1136/bmigh-2020-003901 PMID: 33033054
- de Savigny D, Blanchet K, Adam T,. Applied systems thinking for health systems research: a methodological handbook: McGraw-Hill Education; 2017.
- **83.** Whitaker J, Amoah AS, Dube A, Chirwa L, Munthali B, Rickard RF, et al. Novel application of multifacility process map analysis for rapid injury care health system assessment in Northern Malawi. BMJ Open. 2023; 13(6):e070900. https://doi.org/10.1136/bmjopen-2022-070900 PMID: 37263691
- Nyondo-Mipando AL, Woo Kinshella M-L, Bohne C, Suwedi-Kapesa LC, Salimu S, Banda M, et al. Barriers and enablers of implementing bubble Continuous Positive Airway Pressure (CPAP): Perspectives of health professionals in Malawi. PLoS ONE. 2020; 15(2):e0228915. <u>https://doi.org/10.1371/</u> journal.pone.0228915 PMID: 32053649

- Nyhus HB, Kamara MM. Quality improvement in emergency service delivery: Assessment of knowledge and skills amongst emergency nurses at Connaught Hospital, Sierra Leone. Afr J Emerg Med. 2017; 7(3):113–117. https://doi.org/10.1016/j.afjem.2017.04.002 PMID: 30456121
- 86. Chihana M, Floyd S, Molesworth A, Crampin AC, Kayuni N, Price A, et al. Adult mortality and probable cause of death in rural northern Malawi in the era of HIV treatment. Trop Med Int Health. 2012; 17(8): e74–e83. https://doi.org/10.1111/j.1365-3156.2012.02929.x PMID: 22943382
- 87. McGee K, Sethi D, Peden M, Habibula S. Guidelines for Conducting Community Surveys on Injuries and Violence. World Health Organisation; 2004.
- Holdsworth LM, Safaeinili N, Winget M, Lorenz KA, Lough M, Asch S, et al. Adapting rapid assessment procedures for implementation research using a team-based approach to analysis: a case example of patient quality and safety interventions in the ICU. Implement Sci. 2020; 15(1):12. https://doi.org/10.1186/s13012-020-0972-5 PMID: 32087724
- Palinkas LA, Zatzick D. Rapid Assessment Procedure Informed Clinical Ethnography (RAPICE) in Pragmatic Clinical Trials of Mental Health Services Implementation: Methods and Applied Case Study. Adm Policy Ment Health. 2019; 46(2):255–270. Epub 2018/11/30. https://doi.org/10.1007/ s10488-018-0909-3 PMID: 30488143.
- Sutton-Brown CA. Photovoice: A Methodological Guide. Photogr Cult. 2014; 7(2):169–185. <u>https://doi.org/10.2752/175145214X13999922103165</u>
- Rudolfson N, Gruendl M, Nkurunziza T, Kateera F, Sonderman K, Nihiwacu E, et al. Validating the Global Surgery Geographical Accessibility Indicator: Differences in Modeled Versus Patient-Reported Travel Times. World J Surg. 2020; 44(7):2123–2130. <u>https://doi.org/10.1007/s00268-020-05480-8</u> PMID: 32274536
- Tansley G, Schuurman N, Amram O, Yanchar N. Spatial Access to Emergency Services in Low- and Middle-Income Countries: A GIS-Based Analysis. PLoS ONE. 2015; 10(11):e0141113. <u>https://doi.org/ 10.1371/journal.pone.0141113</u> PMID: 26528911
- 93. Whitaker J, Brunelli G, Van Boeckel TP, Dube A, Amoah AS, Rickard RF, et al. Access to care following injury in Northern Malawi, a comparison of travel time estimates between Geographic Information System and community household reports. Injury. 2022; 53(5):1690–1698. <u>https://doi.org/10.1016/j. injury.2022.02.010</u> PMID: 35153068
- 94. van Duinen AJ, Adde HA, Fredin O, Holmer H, Hagander L, Koroma AP, et al. Travel time and perinatal mortality after emergency caesarean sections: an evaluation of the 2-hour proximity indicator in Sierra Leone. BMJ Glob Health. 2020; 5(12):e003943. <u>https://doi.org/10.1136/bmjgh-2020-003943</u> PMID: 33355267
- 95. Banke-Thomas A, Wong KLM, Ayomoh FI, Giwa-Ayedun RO, Benova L. "In cities, it's not far, but it takes long": comparing estimated and replicated travel times to reach life-saving obstetric care in Lagos, Nigeria. BMJ Glob Health. 2021; 6(1):e004318. https://doi.org/10.1136/bmjgh-2020-004318 PMID: 33495286
- 96. Whitaker J, Chirwa L, Munthali B, Dube A, Amoah AS, Leather AJM, et al. Development and use of clinical vignettes to assess injury care quality in Northern Malawi. Injury. 2021; 52(4):793–805. Epub 2021/01/26. https://doi.org/10.1016/j.injury.2021.01.013 PMID: 33487406.
- Geduld H, Hynes EJC, Wallis LA, Reynolds T. Hospital proximity does not guarantee access to emergency care. Lancet Glob Health. 6(7):e731. https://doi.org/10.1016/S2214-109X(18)30235-3 PMID: 29903374.
- Stewart BT, Gyedu A, Tansley G, Yeboah D, Amponsah-Manu F, Mock C, et al. Orthopaedic Trauma Care Capacity Assessment and Strategic Planning in Ghana: Mapping a Way Forward. J Bone Joint Surg Am. 2016; 98(23):e104. Epub 2016/12/08. https://doi.org/10.2106/JBJS.15.01299 PMID: 27926686; PubMed Central PMCID: PMC5133455.
- Mock C, Nguyen S, Quansah R, Arreola-Risa C, Viradia R, Joshipura M. Evaluation of Trauma Care capabilities in four countries using the WHO-IATSIC Guidelines for Essential Trauma Care. World J Surg. 2006; 30(6):946–956. Epub 2006/06/01. https://doi.org/10.1007/s00268-005-0768-4 PMID: 16736320.
- 100. Brizuela V, Leslie HH, Sharma J, Langer A, Tunçalp Ö. Measuring quality of care for all women and newborns: how do we know if we are doing it right? A review of facility assessment tools. Lancet Glob Health. 2019; 7(5):e624–e632. https://doi.org/10.1016/S2214-109X(19)30033-6 PMID: 30898495
- Wesson HKH, Boikhutso N, Bachani AM, Hofman KJ, Hyder AA. The cost of injury and trauma care in low- and middle-income countries: a review of economic evidence. Health Policy Plan. 2014; 29 (6):795–808. Epub 2013/10/05. https://doi.org/10.1093/heapol/czt064 PMID: 24097794.
- O'Reilly GM, Joshipura M, Cameron PA, Gruen R. Trauma registries in developing countries: a review of the published experience. Injury. 2013; 44(6):713–721. Epub 2013/03/12. <u>https://doi.org/10.1016/j. injury.2013.02.003</u> PMID: 23473265.

- 103. Gidengil CA, Linder JA, Beach S, Setodji CM, Hunter G, Mehrotra A. Using Clinical Vignettes to Assess Quality of Care for Acute Respiratory Infections. Inquiry. 2016; 53:0046958016636531. https://doi.org/10.1177/0046958016636531 PMID: 27098876.
- Christian CS, Gerdtham U-G, Hompashe D, Smith A, Burger R. Measuring Quality Gaps in TB Screening in South Africa Using Standardised Patient Analysis. Int J Environ Res Public Health. 2018; 15 (4):729. https://doi.org/10.3390/ijerph15040729 PMID: 29649095.
- 105. Khan ANS, Karim F, Chowdhury MAK, Zaka N, Manu A, Arifeen SE, et al. Competence of healthcare professionals in diagnosing and managing obstetric complications and conducting neonatal care: a clinical vignette-based assessment in district and subdistrict hospitals in northern Bangladesh. BMJ Open. 2019; 9(8):e028670. https://doi.org/10.1136/bmjopen-2018-028670 PMID: 31427325
- 106. Spreng CP, Ojo IP, Burger NE, Sood N, Peabody JW, Demaria LM. Does stewardship make a difference in the quality of care? Evidence from clinics and pharmacies in Kenya and Ghana. International J Qual Health Care. 2014; 26(4):388–396. https://doi.org/10.1093/intqhc/mzu054 PMID: 24836515
- Peabody JW, Luck J, Glassman P, Jain S, Hansen J, Spell M, et al. Measuring the quality of physician practice by using clinical vignettes: a prospective validation study. Ann Intern Med. 2004; 141 (10):771–780. Epub 2004/11/17. https://doi.org/10.7326/0003-4819-141-10-200411160-00008 PMID: 15545677.
- Roberts I, Shakur H, Edwards P, Yates D, Sandercock P. Trauma care research and the war on uncertainty. BMJ. 2005; 331(7525):1094–1096. <u>https://doi.org/10.1136/bmj.331.7525.1094</u> PMID: 16282383.
- Nordberg E, Oganga H, Kazibwe S, Onyango J. Rapid assessment of an African district health system. Test of a planning tool. Int J Health Plann Manage. 1993; 8(3):219–233. Epub 1993/07/01. https://doi.org/10.1002/hpm.4740080306 PMID: 10134927.
- 110. Mei J, Liang Y, Shi L, Zhao J, Wang Y, Kuang L. The Development and Validation of a Rapid Assessment Tool of Primary Care in China. Biomed Res Int. 2016; 2016:6019603. Epub 2016/02/18. <u>https://doi.org/10.1155/2016/6019603</u> PMID: 26885509; PubMed Central PMCID: PMC4738987.
- 111. Beran D, Yudkin JS, de Courten M. Assessing health systems for type 1 diabetes in sub-Saharan Africa: developing a 'Rapid Assessment Protocol for Insulin Access'. BMC Health Serv Res. 2006; 6:17. Epub 2006/03/01. <u>https://doi.org/10.1186/1472-6963-6-17</u> PMID: <u>16504123</u>; PubMed Central PMCID: PMC1402284.
- 112. Glasgow RE. What Does It Mean to Be Pragmatic? Pragmatic Methods, Measures, and Models to Facilitate Research Translation. Health Educ Behav. 2013; 40(3):257–265. <u>https://doi.org/10.1177/1090198113486805 PMID: 23709579</u>.
- 113. Wang C, Burris MA. Photovoice: concept, methodology, and use for participatory needs assessment. Health Educ Behav. 1997; 24(3):369–387. Epub 1997/06/01. <u>https://doi.org/10.1177/</u> 109019819702400309 PMID: 9158980.
- 114. Tayeb SE, Abdalla S, Mørkve O, Heuch I, Van den Bergh G. Injuries in Khartoum state, the Sudan: a household survey of incidence and risk factors. Int J Inj Contr Saf Promot. 2014; 21(2):144–153. https://doi.org/10.1080/17457300.2013.792283 PMID: 23654301
- 115. Schell LM. Towards the demise of the urban-rural contrast: a research design inadequate to understand urban influences on human biology. Ann Hum Biol. 2018; 45(2):107–109. https://doi.org/10. 1080/03014460.2018.1450445 PMID: 29614926.
- **116.** NIHR Global Health Group on Equitable Access to Quality Health Care for Injured People in Four Low or Middle Income Countries: Equi-injury (NIHR133135). 2022.